

Author's response to reviews

Title: Discriminating lymphomas and reactive lymphadenopathy in lymph node biopsies by gene expression profiling

Authors:

To Ha Loi (t.loi@amr.org.au)
Anna Campain (annac@maths.usyd.edu.au)
Adam Bryant (a.bryant@amr.org.au)
Tim J Molloy (t.molloy@amr.org.au)
Mark Lutherborrow (m.lutherborrow@amr.org.au)
Jennifer Turner (jtur8838@bigpond.net.au)
Yee Hwa Jean Yang (jeany@maths.usyd.edu.au)
David DF Ma (d.ma@amr.org.au)

Version: 5 **Date:** 21 March 2011

Author's response to reviews: see over

Blood, Stem Cell & Cancer Research Unit
Department of Haematology,
St Vincent's Hospital
390 Victoria St
Darlinghurst NSW 2010 Australia

21st March, 2011

Dear Editor of BMC Medical Genomics,

RE: Submission of accepted manuscript 1852645016427570 with formatting changes

Discriminating lymphomas and reactive lymphadenopathy in lymph node biopsies by gene expression profiling

Thank you for accepting the manuscript for publication. As requested, the following formatting changes have been made to this latest version:

- 1) The following text "See Additional files 2 and 3 for the lists of top 10 and 20 classifiers genes distinguishing HL from NHL and FL from DLBCL." has been added to page 10 (paragraph-1) to ensure that additional files are cited in numerical order.
- 2) All track changes in the previous version of the manuscript have been accepted.
- 3) Images in Figure 1 have been further cropped to minimise white space.
- 4) A few typographical and grammatical errors were identified and have been changed.

Please do not hesitate to contact me if there are any problems or oversight.

Sincerely,

David D.F. Ma
Professor of Haematology